Author's response to reviews

Title: Symptom burden in community-dwelling older people with multimorbidity: a cross-sectional study

Authors:

Jeanette Eckerblad (jeanette.eckerblad@liu.se)
Kersti Theander (kersti.theander@kau.se)
Anne Ekdahl (Anne.Ekdahl@lio.se)
Mitra Unosson (mitra.unosson@liu.se)
Ann-Britt Wirehn (Ann-Britt.Wirehn@lio.se)
Anna Milberg (anna.milberg@liu.se)
Barbro Krevers (barbro.krevers@liu.se)
Tiny Jaarsma (tiny.jaarsma@liu.se)

Version: 7 Date: 20 December 2014

Author's response to reviews: see over
Symptom burden in community-dwelling older people with multimorbidity: a cross-sectional study
Jeanette Eckerblad, Kersti Theander, Anne Ekdahl, Mitra Unosson, Ann-Britt Wirehn, Anna Milberg, Barbro Krevers and Tiny Jaarsma

<table>
<thead>
<tr>
<th>Comment</th>
<th>Changes in the text</th>
</tr>
</thead>
<tbody>
<tr>
<td>&quot;Please remove the duplicated text in line 93.&quot;</td>
<td>Thank you so much for noticing this. The changes have been made. Page 5 Line 93.</td>
</tr>
<tr>
<td></td>
<td>The study followed the ethical guidelines given in the declaration of Helsinki and was approved by the regional ethical review board in Linkoping (Dnr. 2011/41-31).</td>
</tr>
</tbody>
</table>

**Best Regards**

On behalf of Professor Tiny Jaarsma and Professor Mitra Unosson.

Jeanette Eckerblad
Department of Social and Welfare Studies,
Faculty of Health Sciences, Linkoping University,
601 74 Norrkoping Sweden